Author's response to reviews

Title: Healthy eating and obesity prevention for preschoolers: a randomised controlled trial

Authors:

Helen Skouteris (helens@deakin.edu.au)
Marita McCabe (marita.mccabe@deakin.edu.au)
Boyd Swinburn (boyd.swinburn@deakin.edu.au)
Briony Hill (briony.hill@deakin.edu.au)

Version: 3 Date: 12 April 2010

Author's response to reviews: see over
Dear Doctor Melissa Norton,

Subject: MS: 1511856226365957

Thank you for provisionally accepting our manuscript entitled “Healthy eating and obesity prevention for preschoolers: a randomised controlled trial”.

I would like to advise of the following changes:

1. The statement “All authors read and approved the final manuscript” has been added to the ‘Authors’ Contributions’ section.
2. Reference 29, under review has been removed from the reference list, and cited in text as “Skouteris H, McCabe M, Swinburn B, Newgreen V, Sacher P, Chadwick P: Parental influence and obesity prevention in preschoolers: A systematic review of interventions, submitted”. The numbered references have been adjusted accordingly. However, we did want reiterate that this paper has been accepted for publication in Obesity, hence, it has undergone peer review.
3. The text ‘study protocol’ has been removed from the title page.
4. The email addresses have been formatted as:
   - MM: marita.mccabe@deakin.edu.au
   - BS: boyd.swinburn@deakin.edu.au
   - BH: brony.hill@deakin.edu.au
5. The ‘Design and Methods’ section has been renamed “Methods/Design”,
6. Full citation details for references 4, 5 and 64 (previously 65) have been provided.
7. Visible vertical lines from Table 1 have been removed.

I look forward to hearing back from you in due course.

Yours sincerely,

Helen

Dr Helen Skouteris
School of Psychology,
Deakin University, Victoria, 3125.
Email: helens@deakin.edu.au